Author's response to reviews

Title: Clear lens phacoemulsification in anterior lenticonus due to Alport syndrome 'a case series'

Authors:

Ghassem Amiraslanzadeh (dr.aslanzadeh@gmail.com)
Davood Gharabaghi (gharabaghid@gmail.com)
Niloofer Naderi (dr_niloofarnaderi@yahoo.com)

Version: 5 Date: 30 August 2007

Author's response to reviews: see over
The management of vision correction in anterior lenticonus due to Alport syndrome was interesting for my colleagues and me. It was indeed a relative urgent indication for refractive lens exchange and I named it “Clear lens phacoemulsification.” In this article we report clear lens phacoemulsification with foldable intraocular lens implantation performed in four eyes of two Alport Syndrome patients with anterior lenticonus. The operations were under general anesthesia and we worked very carefully. All of the four eyes were good after operation and achieved acceptable visual results. We think these reports can be useful in educational fields and the procedure can be a treatment protocol.